Author’s response to reviews

Title: Access to Eye Health Services among Indigenous Australians: an Area Level Analysis

Authors:

Margaret Kelaher (mkelaher@unimelb.edu.au)
Angeline Ferdinand (afer@unimelb.edu.au)
Hugh Taylor (mkelaher@unimelb.edu.au)

Version: 12 Date: 4 September 2012

Author’s response to reviews:

All outstanding comments have been addressed

- Background: This sentence needs a reference: “The rate of low vision in Indigenous adults is 2.8 times rates in the general population and the rate of blindness 6.2 times higher than general population rates?.

Added
- Please spell out ACT at first use (end of inpatient section).
Done
- Please reference the SEIFA website.

- I still find the discussion to be narrow. The authors should at least discuss how their results on eye health access fit with other research on health care access in indigenous populations of Australia.

Further discussion has been added in this and the previous revision.

Thank you for providing the descriptive data that was requested by the reviewer however please also provide some comment on the reviewer's concerns about the ability to properly adjust for remoteness if the areas with very high Indigenous populations are all remote.

Table 1 clarifies that not all areas with high Indigenous populations are remote.